Author’s response to reviews

Title: LINGUISTIC-CULTURAL VALIDATION OF THE ORAL HEALTH ASSESSMENT TOOL (OHAT) FOR THE ITALIAN CONTEXT

Authors:
Stefano Finotto (sfinotto@unimore.it)
Giorgia Bertolini (giorgia.bertolini@ausl.re.it)
Riccarda Camellini (rcamellini@unimore.it)
Rita Fantelli (rfantelli@unimore.it)
Debora Formisano (debona.formisano@ausl.re.it)
Maria Grazia Macchioni (macchionim@unimore.it)
Daniela Mecugni (daniele.mecugni@unimore.it)

Version: 1 Date: 27 Sep 2018

Author’s response to reviews:

1. Ethics approval wasn’t necessary according with national guidelines and regulation as reported on the manuscript (ref. Lines from 388 to 392).

2. Abstract’s format has been reformulated following the subheadings you requested (ref. Lines of the manuscript from 39 to 67).

3. List of abbreviation has been add at the end of ‘Conclusions’ section as you requested (ref. Lines of the manuscript from 375 to 385).

4. Clean version of the manuscript is submitted.